Author's response to reviews

Title: Stages of change: A qualitative study on the implementation of a perinatal audit program in South Africa

Authors:

María Belizán (maria.belizan@gmail.com)
Anne-Marie Bergh (anne-marie.bergh@up.ac.za)
Carolé Cilliers (22573143@nwu.ac.za)
Robert C Pattinson (robert.pattinson@up.ac.za)
Anna Voce (voceas@ukzn.ac.za)

Version: 14 Date: 27 September 2011

Author's response to reviews:

I have made a few small typographical changes to make conventions consistent:
(1) "backup/s": all made "back-up/s"
(2) "and or": all made "and/or"
(3) "data" plus verb: all made singular
(4) Stage 5 of table: three of the sentences beginning with an "H" had the "H" with a strikethrough line - they are now corrected

Many thanks
Anne-Marie Bergh